Antibiotics for lower respiratory tract infection in children presenting in primary care: ARTIC-PC RCT

Paul Little,^{1*} Nick A Francis,¹ Beth Stuart,¹ Gilly O'Reilly,¹ Natalie Thompson,¹ Taeko Becque,¹ Alastair D Hay,² Kay Wang,³ Michael Sharland,⁴ Anthony Harnden,³ Guiqing Yao,⁵ James Raftery,¹ Shihua Zhu,¹ Joseph Little,¹ Charlotte Hookham,¹ Kate Rowley,² Joanne Euden,⁶ Kim Harman,¹ Samuel Coenen,⁷ Robert C Read,^{8,9} Catherine Woods,¹ Christopher C Butler,³ Saul N Faust,^{8,9} Geraldine Leydon,¹ Mandy Wan,¹⁰ Kerenza Hood,⁶ Jane Whitehurst,¹¹ Samantha Richards-Hall,¹² Peter Smith,¹³ Michael Thomas,¹ Michael Moore¹ and Theo Verheij^{1,14}

- ¹Primary Care Population Sciences and Medical Education Unit, University of Southampton, Southampton, UK
- ²Centre for Academic Primary Care, Bristol Medical School, Population Health Sciences, University of Bristol, Bristol, UK
- ³Nuffield Department of Primary Care Health Sciences, University of Oxford, Oxford, UK
- ⁴Institute of Infection and Immunity, St George's University, London, UK
- ⁵Biostatistics Research Group, Department of Health Sciences, College of Life Sciences, University of Leicester, Leicester, UK
- ⁶Centre for Trials Research, College of Biomedical and Life Sciences, Cardiff University, Cardiff, UK
- ⁷Department of Family Medicine & Population Health and Vaccine & Infectious Disease Institute, University of Antwerp, Antwerp, Belgium
- ⁸National Institute for Health and Care Research (NIHR) Southampton Clinical Research Facility and Biomedical Research Centre, University Hospital Southampton NHS Foundation Trust, Southampton, UK
- ⁹Faculty of Medicine and Institute for Life Sciences, University of Southampton, Southampton, UK
- ¹⁰Evelina Pharmacy, Guy's and St Thomas' NHS Foundation Trust, London, UK
- ¹¹National Institute for Health and Care Research (NIHR) Applied Research Collaboration West Midlands, Coventry, UK

- ¹²Southampton Primary Care Research Centre, Primary Care Population Sciences and Medical Education Unit, University of Southampton, Southampton, UK
- ¹³Southampton Statistical Sciences Research Institute, University of Southampton, Southampton, UK
- ¹⁴Julius Centre for Health Sciences and Primary Care, University Medical Centre Utrecht, Utrecht, the Netherlands

*Corresponding author p.little@soton.ac.uk

Declared competing interests of authors: Theo Verheij reports grants from the European Union and The Netherlands Organisation for Health Research and Development during the conduct of the study, and grants from Abbott Laboratories (Chicago, IL, USA), Becton, Dickinson and Company (Franklin Lakes, NJ, USA), bioMérieux (Marcy-l'Étoile, France) and Janssen Pharmaceuticals (Beerse, Belgium) outside the submitted work. Paul Little was a member of the National Institute for Health and Care Research (NIHR) Journals Library Board (2011–17). Kerenza Hood is a member of the NIHR Health Technology Assessment (HTA) General Committee and the HTA Funding Strategy Group. Saul N Faust is a member of the RECOVERY Trial Steering Committee, the HTA Commissioning Board and other NIHR national groups, and reports funds to his institution but no personal payments for multiple commercial contracts, advisory board participation and presentations [Pfizer (New York, NY, USA), AstraZeneca (Cambridge, UK), MedImmune (Gaithersburg, MD, USA), Sanofi (Paris, France), CSL Segirus (Maidenhead, UK) and Merck Group (Darmstadt, Germany)]. Robert C Read reports that he is NIHR Biomedical Research Centre Director and editor-in-chief of Journal of Infection. James Raftery reports membership of the NIHR HTA and Efficacy and Mechanism Evaluation Editorial Board in a paid role. Nick A Francis reports a research grant from Synairgen plc (Southampton, UK) for COVID-19 research, non-financial support for a NIHR HTA trial on reducing antibiotics for COPD from Abbott Laboratories, a speaker's fee from Abbott Laboratories, and stock/stock options from Abbott Laboratories and Synairgen plc.

Published June 2023 DOI: 10.3310/DGBV3199

Plain language summary

Antibiotics for lower respiratory tract infection in children presenting in primary care: ARTIC-PC RCT

Health Technology Assessment 2023; Vol. 27: No. 9 DOI: 10.3310/DGBV3199

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain language summary

Background

Children are commonly prescribed antibiotics for chest infections, but such infections are becoming resistant to antibiotics, and it is not clear if antibiotics work in treating them.

Methods

A total of 432 children who saw their general practitioner with a chest infection were given either an antibiotic (amoxicillin) or a placebo (no antibiotic) for 7 days. Symptom diaries documented the infection's duration and its side effects. Children not in the placebo study were able to participate in another study that documented the same outcomes (an 'observational study'). We interviewed parents, doctors and nurses about their observations and concerns. Our patient and public involvement and engagement work with parents indicated that a 3-day symptom reduction was required to justify giving antibiotics.

Results

After seeing the doctor, parents whose children received antibiotics rated infective symptoms as moderately bad or worse for 5 days, and parents whose children received the placebo rated these for 6 days. Side effects and complications were similar in the two groups. Findings were similar when including the results of the observational study, and for children in whose chest the doctor could hear wheeze or rattles; who had fever; who were rated by the doctor as more unwell, who were short of breath, or who had had bacteria detected in the throat. The costs to the NHS per child were similar (antibiotics, £29; placebo, £26), and the wider costs to society were the same (antibiotics, £33; placebo, £33).

Parents found it difficult to interpret their child's symptoms, and commonly used the sound of the cough to judge severity. Parents commonly consulted to receive an examination and reassurance, and accepted that antibiotics should be used only when 'necessary'. Clinicians noted a reduction in parents' expectations for antibiotics.

Conclusion

Amoxicillin for chest infections in children is unlikely to be effective. General practitioners should support parents to self-manage at home and give clear communication about when and how to seek medical help if they continue to be concerned.

Health Technology Assessment

ISSN 1366-5278 (Print)

ISSN 2046-4924 (Online)

Impact factor: 4.014

Launched in 1997, *Health Technology Assessment* (HTA) has an impact factor of 4.014 and is ranked 27th (out of 108 titles) in the 'Health Care Sciences & Services' category of the Clarivate 2021 Journal Citation Reports (Science Edition). It is also indexed by MEDLINE, CINAHL (EBSCO Information Services, Ipswich, MA, USA), Embase (Elsevier, Amsterdam, the Netherlands), NCBI Bookshelf, DOAJ, Europe PMC, the Cochrane Library (John Wiley & Sons, Inc., Hoboken, NJ, USA), INAHTA, the British Nursing Index (ProQuest LLC, Ann Arbor, MI, USA), Ulrichsweb™ (ProQuest LLC, Ann Arbor, MI, USA) and the Science Citation Index Expanded™ (Clarivate™, Philadelphia, PA, USA).

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full HTA archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hta.

Criteria for inclusion in the Health Technology Assessment journal

Reports are published in *Health Technology Assessment* (HTÅ) if (1) they have resulted from work for the HTA programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Reviews in *Health Technology Assessment* are termed 'systematic' when the account of the search appraisal and synthesis methods (to minimise biases and random errors) would, in theory, permit the replication of the review by others.

HTA programme

Health Technology Assessment (HTA) research is undertaken where some evidence already exists to show that a technology can be effective and this needs to be compared to the current standard intervention to see which works best. Research can evaluate any intervention used in the treatment, prevention or diagnosis of disease, provided the study outcomes lead to findings that have the potential to be of direct benefit to NHS patients. Technologies in this context mean any method used to promote health; prevent and treat disease; and improve rehabilitation or long-term care. They are not confined to new drugs and include any intervention used in the treatment, prevention or diagnosis of disease.

The journal is indexed in NHS Evidence via its abstracts included in MEDLINE and its Technology Assessment Reports inform National Institute for Health and Care Excellence (NICE) guidance. HTA research is also an important source of evidence for National Screening Committee (NSC) policy decisions.

This report

The research reported in this issue of the journal was funded by the HTA programme as project number 13/34/64. The contractual start date was in July 2016. The draft report began editorial review in April 2021 and was accepted for publication in February 2022. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HTA editors and publisher have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the draft document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health and Care Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, the HTA programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the NHS, these of the authors, those of the NHS, the NIHR, the HTA programme or the Department of Health and Social Care.

Copyright © 2023 Little *et al*. This work was produced by Little *et al*. under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This is an Open Access publication distributed under the terms of the Creative Commons Attribution CC BY 4.0 licence, which permits unrestricted use, distribution, reproduction and adaption in any medium and for any purpose provided that it is properly attributed. See: https://creativecommons.org/licenses/by/4.0/. For attribution the title, original author(s), the publication source – NIHR Journals Library, and the DOI of the publication must be cited.

Published by NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress, final files produced by Newgen Digitalworks Pvt Ltd, Chennai, India (www.newgen.co).

NIHR Journals Library Editor-in-Chief

Dr Cat Chatfield Director of Health Services Research UK

NIHR Journals Library Editors

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HSDR, PGfAR, PHR journals) and Editorin-Chief of HSDR, PGfAR, PHR journals

Dr Peter Davidson Interim Chair of HTA and EME Editorial Board. Consultant Advisor, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Professor Matthias Beck Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Consultant in Public Health, Delta Public Health Consulting Ltd, UK

Ms Tara Lamont Senior Adviser, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Dr Catriona McDaid Reader in Trials, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Emeritus Professor of Wellbeing Research, University of Winchester, UK

Professor James Raftery Professor of Health Technology Assessment, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Dr Rob Riemsma Consultant Advisor, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Professor Helen Roberts Professor of Child Health Research, Child and Adolescent Mental Health, Palliative Care and Paediatrics Unit, Population Policy and Practice Programme, UCL Great Ormond Street Institute of Child Health, London, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk